Author's response to reviews

Title: A randomized controlled trial testing the effectiveness of a universal school-based depression prevention program 'Op Volle Kracht' in the Netherlands

Authors:

Yuli R Tak (y.tak@pwo.ru.nl)
Rinka MP Van Zundert (R.vanZundert@pwo.ru.nl)
Rowella CWM Kuijpers (r.kuijpers@pwo.ru.nl)
Boukje S Van Vlokhoven (bvlokhoven@trimbos.nl)
Hettie FW Rensink (hrensink@trimbos.nl)
Rutger CME Engels (r.engels@bsi.ru.nl)

Version: 6 Date: 9 January 2012

Author's response to reviews:

Dear Sir/Madam,

In this revision I made the suggested changes to my references. First, I replaced reference 62 for the reference in which the Dutch translation of the scale is validly used. Second, I replaced reference 65 for the article about the IM-P scale that is currently been submitted. Third, I removed reference 70.

Kind regards,

Yuli Tak